

# **OPEN ACCESS**

EDITED BY Victoria Team, Monash University, Australia

REVIEWED BY
Adefunke DadeMatthews,
Auburn University, United States
Michelle Villeneuve,
The University of Sydney, Australia

\*CORRESPONDENCE
Phillip Harniess

☑ p.harniess@exeter.ac.uk

RECEIVED 08 June 2024 ACCEPTED 09 September 2024 PUBLISHED 30 September 2024

### CITATION

Harniess P, McGlinchey C, McDonald A, Boyle F, Garrood A, Logan S, Morris C and Borek AJ (2024) Ensuring equitable access, engagement and ability of socially and ethnically diverse participants to benefit from health promotion programmes: a qualitative study with parent carers of disabled children. *Front. Public Health* 12:1445879. doi: 10.3389/fpubh.2024.1445879

# COPYRIGHT

© 2024 Harniess, McGlinchey, McDonald, Boyle, Garrood, Logan, Morris and Borek. This is an open-access article distributed under the terms of the Creative Commons Attribution License (CC BY). The use, distribution or reproduction in other forums is permitted, provided the original author(s) and the copyright owner(s) are credited and that the original publication in this journal is cited, in accordance with accepted academic practice. No use, distribution or reproduction is permitted which does not comply with these terms.

# Ensuring equitable access, engagement and ability of socially and ethnically diverse participants to benefit from health promotion programmes: a qualitative study with parent carers of disabled children

Phillip Harniess<sup>1\*</sup>, Caomhan McGlinchey<sup>1</sup>, Annabel McDonald<sup>1</sup>, Fleur Boyle<sup>1</sup>, Alice Garrood<sup>1</sup>, Stuart Logan<sup>1</sup>, Christopher Morris<sup>1</sup> and Aleksandra J. Borek<sup>2</sup>

<sup>1</sup>Peninsula Childhood Disability Research Unit (PenCRU) and NIHR Applied Research Collaboration South West Peninsula (PenARC), University of Exeter Medical School, University of Exeter, Exeter, United Kingdom, <sup>2</sup>Nuffield Department of Primary Care Health Sciences, Medical Sciences Division, University of Oxford, Radcliffe Observatory Quarter, Oxford, United Kingdom

**Background:** Equity is fundamental to health promotion programmes. However, unintentional or unseen barriers may exist for some underserved groups. We aimed to identify how to ensure equitable access and engagement for diverse parent carers of disabled children to benefit from health promotion programmes.

**Methods:** We purposively sampled parent carers with potentially intersecting characteristics including those who self-identified as from ethnic groups, whose children were educated other than at school, with sensory impairments, or neurodiversity, and fathers. Participants were recruited through local and national organisations and parent carer networks. Data collection involved semi-structured individual interviews, which were transcribed verbatim and analysed thematically and iteratively alongside data collection. Core researchers performed early analysis independently, followed by research team and advisory group cross-validation.

**Results:** Thirty-six parent carers with intersecting characteristics across the sampled backgrounds participated. We identified various perceived barriers around finding out about, attending and engaging with health programmes. We organised the findings into five themes focused on concepts capturing challenges and potential solutions to contextual barriers to access and participation in health programmes. (i) Reach—judiciously using targeted and universal strategies to ensure equitable distribution; (ii) Credibility—demonstrating trustworthiness of those advertising and/or delivering the programme; (iii) Opportunity—ensuring that the programme is seen as fulfilling a relevant need; (iv) Reservations—addressing barriers of readiness to participate; and (v) Optimisation—tailoring to improve the inclusivity of the programme delivery.

**Conclusion:** We identified modifiable factors that impede members of some social groups from engaging with, and benefiting from, health promotion programmes, and potential solutions. We advocate a multifaceted approach is required from outreach to delivery, tailored to be mindful of extant diverse needs of parent

carers in underserved communities. We catalogue key considerations to inform implementation strategies to optimise equity in health programmes for parent carers. The implications are likely transferable to other interventions and contexts.

KEYWORDS

behaviour change interventions, health inequality, wellbeing, caregivers, implementation science, interviews, social determinants of health, community interventions

# Introduction

Public health promotion interventions are "planned actions to prevent or reduce a particular health problem, or the determinants of the problem, in a defined population," [p. 520, (1)] and so require intentionality in their design and implementation. Despite good intentions, there is a risk that these interventions might reproduce or worsen inequalities by primarily favouring individuals with greater advantages (2). Therefore, investigating how interpersonal (e.g., family), social (e.g., ethnicity) and wider contextual factors influence health behaviour change interventions is critical to informing programme design and delivery to promote equity (3-6). Accessibility and acceptability may also be unequal for some groups compared to others, which has implications for participant recruitment and intervention evaluation (4, 7, 8). Some health promotion and behaviour change interventions have shown promise at reducing inequity through contextual adaptation at individual and community levels with different groups (6, 9, 10). However, there are notable research gaps; (i) in understanding how to reach and engage underserved groups in health research (7, 8) and; (ii) in accounting for how social determinants influence interventions and outcomes across different social contexts (6).

The concept of health equity has been defined as the "highest level of health for all people," "opportunity for all," and "absence of disparities" (11). Proportionate universalism proposes that "greater intensity of action is likely to be needed for those with greater social and economic disadvantage" [p. 16, (5)]. Descriptions of populations with greater social and economic disadvantage have drawn debate. Descriptors such as "underserved" or "equity deserving," as opposed to "vulnerable" or "equity-seeking," underline the principle that marginalised groups deserve health equity as a right, and "seeking" should not be their burden (12). We use "underserved," where it is defined as:

"A group that is less well represented in research than would be desirable from population prevalence and healthcare burden." (13)

In comparison to other parents, parent carers of disabled children are more likely to experience compromised physical and mental health and wellbeing (14–17). Additionally, parent carers are more likely to derive from disadvantaged groups (18, 19). While arguably all parent carers have underserved health needs, there may be structural inequalities affecting some groups more than others, which are associated with higher risk (18–21). These inequalities map broadly onto well-established social determinants of population

health (5), whereby personal, psychosocial and economic factors interact to shape and exacerbate health and wellbeing problems (22). Marmot (5), indicates that "these factors are influenced by social position, itself shaped by education, occupation, income, gender, ethnicity and race" (p. 16).

A health and wellbeing promotion programme uses systematic design and actions to empower individuals and communities to address health determinants with the aim of improving health and wellbeing (23). Henceforth we will refer simply to "health programmes." A range of parent carer-focused health programmes have been developed to improve health outcomes for parent carers and their children (24-27). Although different in their design and approach, they have a shared challenge of reaching underserved parent carers who would benefit from such interventions. We co-created Healthy Parent Carers (HPC) as a community-based programme that aims to improve the health and wellbeing of parents of disabled children (Box 1) (24, 28-30). The HPC programme is predominantly delivered by third-sector organisations on behalf of Local Authorities. Third-sector organisations refer to those outside of the state or market, incorporating diverse groups such as charities, nongovernmental organisations and social enterprises (31). Within the UK, third-sector organisations are often commissioned to deliver health programmes.

The aim of our study was to identify how to help ensure equitable access and engagement for socially and ethnically diverse parents of disabled children to benefit from parent carer-focused health programmes.

Regarding terminology, we acknowledge debate around differing preferences for referring to "disabled children" or "children with disability." We use "disabled children," preferred by advocates of the social model of disability, because the child/person is disabled by society or the environment (32, 33). "Parent carer" is a UK legal term derived from the Children and Families Act (2014) to describe parents who are primary adult caregivers of disabled children and young people up to 25 years (34). These children and young people are assigned Special Educational Needs and Disability (SEND) rights for inclusive education (35). We have adopted the term "Global Majority" as a more appropriate descriptor of persons often labelled as "ethnic minorities" as this term challenges "the debilitating implications of being racialised as minorities" through a Western normative lens [p. 22, (36)]. Further, we were conscious to avoid applying "master" categories of social position, such as race/ethnicity, to be proxy for other characteristics, such as poverty or educational status. Intersectionality considers how various aspects of one's identity (e.g., race, gender or educational status) interact to create unique experiences of privilege or disadvantage (37) that may also compound barriers to health interventions, such as those relevant for parent carers in the

BOX 1 Overview of HPC programme.

### Background:

The Healthy Parent Carers (HPC) programme aims to promote parent carers' engagement with health promoting behaviours to improve resilience and overall health and wellbeing.

Co-created by parent carers and researchers, it is based upon behaviour change techniques (e.g., goal setting) and a set of universal and evidence-based actions associated with health and wellbeing: CLANGERS, which stand for Connect, Learn, be Active, take Notice, Give, Eat well, Relax and Sleep, an extension of "the five ways to wellbeing" (28).

## Delivery:

- Groups consist of 6-12 parent carers
- In-person or online via videoconferencing (e.g., Zoom)
- Trained Lead and Assistant Facilitators, who are also parent carers, deliver the programme
- 12 modules with two-hour sessions covering CLANGERS content and behaviour change techniques, such as goal setting and self-monitoring
- Facilitated group-based activities utilising health-related information and resources, via printed or online materials and videos

### Setting:

To date UK based

Spans different geographical regions from urban to more rural

# Summary of programme theory:

The peer-led group-based programme is intended to facilitate change by providing opportunities for and prompting: social and emotional (peer) support, valuing the shared social identity as parent carers who want to look after their health, sharing experiences, and embedding practice of health-promoting behaviours.

More details, including the intervention and implementation logic models, are published elsewhere (24, 28, 30).

context of this study. Therefore, our study sought to understand the context of inequity in parent carers' lives relating to their health and wellbeing and access to programmes in all its complexity, through intersectionality (38).

# Methods and materials

# Public and other stakeholder involvement

We carried out Patient and Public Involvement (PPI) through our Family Faculty group facilitated by a Family Involvement Co-ordinator (AMcD) who is a parent carer. The Family Faculty is made up of families of disabled children in the UK who are interested and involved in work carried out by the research group (39). For this study, we convened a study-specific PPI working group with volunteers from members of the Family Faculty from diverse backgrounds, many of whom had experience of the HPC programme through its co-design process, or as attendees. The PPI group advised on all stages of the research. The Guidance for Reporting Involvement of Patients and the Public (GRIPP-2) short form (40) is used to describe the collaboration with our PPI group for this study (Table 1).

In addition, we consulted with two groups of facilitators of the HPC programme regarding the findings. This enabled further sense-checking and an immediate opportunity to consider the delivery and implementation strategies.

# Participant sampling

All participants were parent carers of disabled children living in the UK. We used purposive maximum variation sampling for parent carers across different underserved groups who had experience of the HPC programme (as facilitators or participants) or no experience. Sampling was informed by our PPI group which itself included parent carers from diverse backgrounds, as well as by UK national guidance on reducing healthcare inequalities (41). We focused on groups who would provide insight into different aspects of inequity in this area, including: ethnically diverse, fathers, parents with sensory impairments and parents whose children are educated at home. This latter group includes parents of children Educated Other Than At School (EOTAS) (42) and those who are Electively Home Educated (EHE) (43). We recorded other relevant intersecting characteristics, including neurodiversity, Indices of Multiple Deprivation (44), educational status, single/co-parenting status and child history. While we were cognisant of intersectionality as important in equity research (45), sampling of all potentially possible intersectional permutations was unfeasible. As we analysed data and recruited participants concurrently, we identified specific groups who continued to provide novel insights even after our initial analysis, akin to theoretical sampling (46). Therefore, we sampled proportionately more participants from ethnically diverse backgrounds.

# Recruitment strategy

Our recruitment strategy was multifaceted. We posted adverts on social media platforms with invitations including a link to the study website with participant information and an online expression of interest form. The expression of interest form asked potential participants about their characteristics to allow for purposive sampling. All potential participants who had expressed their interest were contacted by telephone or e-mail and screened. Eligibility was assessed against the basic parent carer information and the fit with the target sample groups. We informed those who were ineligible by email or phone.

To ensure recruitment from underserved groups, we took additional steps. Ethnically diverse members of our PPI group approached parent carers within their communities about the study. We collaborated with local and national third-sector organisations for the target sample groups, who shared the study adverts across their networks. In four cases during the recruitment process, prospective participants were offered a chaperone while being interviewed. The offer of a chaperone is considered respectful in many communities, and those who elect to have chaperones feel more relaxed, supported, and safe (47). In cases where this offer was accepted, a chaperone was identified. In three cases the chaperone was a female member of our PPI group, she was from a Pakistani background and was familiar to the participants; in the fourth case, a representative from a national charity was requested by the parent carer with a visual impairment to support them as a chaperone because they had a strong pre-existing working relationship.

 ${\sf TABLE\,1\ Public\ and\ patient\ involvement\ in\ this\ research\ described\ using\ the\ {\sf GRIPP-2\ short\ form}.}$ 

Topic	Item
Aims Report the aim of PPI in the study	The diverse group of parent carer PPI contributors were encouraged to bring their lived experience and expertise with the aims to:  Provide a focus on the most meaningful aspects/factors of equity in parent carer-focused health programmes  Contribute to the study sampling approach by identifying and prioritising target groups of parent carers who might experience inequity  Provide advice and practical support for the recruitment strategy, e.g., access to diverse parent carer networks  Pilot and feedback on the interview topic guide  Challenge the researchers' implicit biases and assumptions with a particular emphasis on the interpretation of findings  Consider the implications of findings for implementation strategies locally and for broader research/practice in this field  Support sharing of findings  Further stakeholder meetings were held with HPC programme facilitators to further consider the implications of the findings for programme delivery and implementation.
Methods Provide a clear description of the methods used for PPI in the study	Over the course of the study 10 parent carers were involved in the PPI group meetings, six of whom had previous experience of the HPC programme as participants, one was a trained facilitator; and the others did not have direct experience of HPC. PPI group members were from diverse ethnic backgrounds and lived in different urban and more rural settings across the UK; one member was a father. Group meetings were co-designed and co-facilitated between two researchers and the Family Involvement Coordinator (who is also a parent carer) to support input from parent carers relating to all aspects of the study. Five group online meetings were convened and focused around the stated aims:  Consensus discussions supported the prioritisation of target groups and sampling approach.  Parents were encouraged to advise and provide support connecting the researchers to other organisations and networks of target underserved parent carers groups. We followed up individually with those who offered to support around this activity.  One PPI group member went further by adopting a chaperone role to parent carers from a Muslim background in three interviews.  The interview topic guide was piloted and cognitively tested during one PPI meeting and feedback was given specifically relating to wording and interpretation of questions and areas for potential further probing.  As data began to be collected, emerging analysis was shared with the group for discussion through vignettes to sense-check and challenge the assumptions of the researchers to build a more robust representation and interpretation of the data.  PPI members supported the strategy around the application of the findings into practice, for example suggesting key stakeholders to involve in the next stage of implementation.  PPI members were also involved outside of the group meetings in reviewing study documents, such as the participant information sheet, study advert and consent form and later in the development of the plain English summary of the findings for wider sharing.
Results Outcomes—report the results of PPI in the study, including both positive and negative outcome	implications of the findings for programme delivery and implementation.  Parallel engagement work was undertaken to support our awareness and the recruitment strategy, involving key organisations: The Fatherhood Institute, DadMatters, Look UK, Tower Hamlets Opportunity Group, The Esteem Team.  PPI strengthened the study in multiple ways:  - Directly enabled recruitment of parents from underserved communities, e.g., one member, a British Pakistani mother, became a gatekeeper inviting eight potential participants, five of whom decided to take part.  - A PPI group member chaperoned for Muslim mothers, where requested, and directly supported some parents with low digital literacy to access the video conference call for interviews.  - The PPI group advised the researchers directly around cultural sensitivities when approaching and engaging with members of their ethnic community.  - Other groups, e.g., charities provided similar strategic advice relating to recruitment.  - The study topic guide was developed with the PPI group, who advised on the use of language and tested it through pilot group interviews.  - The data analysis was presented and discussed with the PPI group acting as critical friends. Overall, they reported a resonance with the findings and felt they were sensitively developed.  - They provided direct advice on the relevance of the findings and implementation ideas for parent carer health programmes and HPC specifically.
Discussion and conclusions Outcomes—Comment on the extent to which PPI influenced the study overall	As discussed above, the PPI had a positive and direct influence on the study; specifically, it supported success around recruitment, the collection of rich data and the development of credible and resonant findings and identification of practical implications.

(Continued)

TABLE 1 (Continued)

Topic	Item	
Reflections/critical perspective	The data collection via interviews occurred online for feasibility and pragmatic reasons. When engaging with different	
Comment critically on the study,	communities, a multipronged approach was used to build trust and relationships. However, we had limitations on being able to go	
reflecting on the things that went well	to meet with different groups face-to-face. Meeting face-to-face may have optimised relationship building in certain communities	
and those that did not, so others can	and the participation of more underserved parent carers. Nevertheless, our strategy did still enable effective involvement of many	
learn from this experience	parent carers who we would not have been able to access otherwise.	
	Further engagement work is needed with the PPI group to build on the implementation strategy and its application into practice.	

All participants provided consent and received a £35 voucher following interview.

# Data collection

We conducted one-to-one semi-structured in-depth interviews via video conference calls from September 2023 to February 2024. Interviews were conducted by one of two researchers (CMcG/PH) and lasted approximately one hour. A semi-structured topic guide was created and piloted with the PPI group (Supplementary Document). It included open questions, followed by more specific prompts, about accessing and engaging with health programmes. Within the framing of their social identity (e.g., "as a father ..."), we encouraged participants to reflect on factors that enabled or impeded their access and engagement with such programmes. Interviewees who had attended the HPC programme were asked to reflect on these factors both before and after joining the course and facilitators in relation to its delivery. Interviewees who had no experience of HPC were asked to consider factors relevant to them and their community when thinking about other hypothetical or experienced parent-carer programmes. Interviews were audiorecorded and transcribed verbatim.

# Data analysis

We used reflexive thematic analysis methods (48). Our process involved recording written summaries of each interview, which we used during weekly team discussions alongside data collection to enable a responsive approach to subsequent recruitment, interview probing and analysis. Verbatim transcripts were uploaded to NVivo v.14 software for coding. To develop a codebook, three researchers independently coded the same sub-sample of transcripts (CMcG and PH n=5 each, AB n=2). Each researcher arranged the codes into candidate categories. The categories and codes were compared and discussed and integrated into one combined codebook. It was then used by CMcG and PH in subsequent analyses of transcripts. Where new and relevant phenomena were observed, additional discussions supported the further development of the coding framework. Findings were regularly reviewed with the research team and periodically with the study PPI and stakeholder groups.

The researchers conducting the analysis included two White British males and a White European female from diverse professional and academic backgrounds (educational psychology, physiotherapy and sociology). To remain cognisant of our positionality during interviews and within the analysis and ensure rigour, we used reflexive practices such as writing memos, (peer) supervision and discussing differences in coding interpretations. In addition, triangulation was vital with the PPI and stakeholder groups, which included those from underserved groups, to increase the validity of findings (49, 50).

# **Ethics approval**

The study procedures were approved by the University of Exeter Medical School Research Ethics Committee (UEMS REC ID: 525009).

# Results

Thirty-six parent carers were interviewed; 20 had never attended a parent carer-focused health programme, 10 had previously attended the HPC programme, and six were Lead or Assistant Facilitators on the HPC programme. There were 16 from Global Majority groups, 12 were fathers, seven were parents of children who were EOTAS and four parents had sensory impairment (Table 2).

We constructed five themes relevant to enhancing equity in access and engagement with parent carer-focused health programmes, which focused on the central concepts: "Reach," "Credibility," "Opportunity," "Reservations" and "Optimisation" (Table 3). We elucidate upon each theme using supporting quotes, attributed with initials to indicate whether the participants were HPC programme attendees (A), non-attendees (NA) or facilitators (F) and other relevant characteristics. Our findings are presented for the whole sample and, where relevant, we note barriers that linked with specific characteristics (e.g., ethnicity, sensory impairment).

# Reach: judiciously using targeted and universal strategies to ensure equitable distribution

A primary challenge acknowledged across groups, and core to health inequity, was reaching those less or not engaged with health programmes, rather than "engaging with people who are already engaged" (A3, British Asian, father). "Reach" accounts for the intentional and informed effort to communicate with underserved populations because it is recognised that some parent carers are not connected into networks and places where programmes might be advertised.

TABLE 2 Participant characteristics (n = 36).

		N	%
Purposively sampled	Global Majority	16	44.4
characteristics	background		
	EOTAS or EHE*	7	19.4
	Fathers	12	33.3
	Sensory impairment	4	11.1
Additional relevant characteristic	Neurodiverse (ASC/ ADHD**)	4	11.1
Parent carer mean age— years (Range)		47 (33–60)	
Parent carer sex	Female	24	66.7
	Male	12	33.3
Partner at home	Yes	23	63.9
	No	13	36.1
Parent education	Degree or above	27	75.0
	Educated to 19 years old	3	8.3
	Educated to 16 or below	6	16.7
Child's main diagnosis	ASC/ADHD*	31	86.0
	Learning disability	4	9.7
	Physical neurodisability (e.g., cerebral palsy)	6	13.0
	Visual or hearing impairment	5	12.2
Mean child age—years (Range)		12.3 (3–20)	
Deprivation in area of	1 (most deprived)	5	15.1
residence (Indices of	2	15	45.4
multiple	3	3	9.0
deprivation***—IMD; quintiles)	4	3	9.0
	5 (least deprived)	7	21.2

<sup>\*</sup>Educated Other Than At School (EOTAS) | Electively Home Educated (EHE).

"... people who are not part of any of the networks ... how likely they are to sign up for it. I do not know." (F36, British Bengali mother)

In more patriarchal cultures, it was explained that "you need to convince the husband as well" (NA20, Asian mother). Particularly where English was a second language for the mother, fathers were more likely to source relevant information (e.g., via school or the local mosque/masjid) and pass it on to their partners.

Participants, in reference to their group, highlighted that it was commonplace that there were parents unfamiliar with the "system" who do not know where to look or how to ask for support, including such health programmes.

"... how are people supposed to get information? ... how do they know where to go in the first place though? [Laughs] ... the

demographics of this area, there's people that do not know the system, they do not speak the language, there's no advocacy services." (NA14, British Pakistani mother)

This scenario might represent an inequity in SEND cultural capital, whereby leveraging support for their child and family is constrained, and a barrier to finding out about parent carer-focused programmes.

Participants discussed contrasting universal or targeted advertising strategies. A universal approach taps into existing statutory structures and services with "blanket" coverage and contact with parent carers in places, such as Special Educational Needs schools, children's centres and the health sector. It was seen to have comprehensive reach. In particular "the diagnostic services" for the child (e.g., paediatricians and child development centres) were identified as potentially strategic distributors because "that's when parents are seeking help at that point." One parent argued that going through statutory bodies was most equitable because it avoided the "inconsistent," "disjointed" and "patchwork" coverage of third sector organisations. Social media was also advocated for as an advertising strategy.

However, this universal strategy was seen as less effective at reaching underserved groups. Single fathers reported being excluded from their child's Special Educational Needs (SEN) school email lists, while parents of EOTAS children stated that they "end up disappearing" by no longer having access to conventional information shared through schools. Organisations, such as local authorities, that might control distribution were criticised as "sending emails blindly" to mailing lists that were inherently exclusive, while simultaneously not working hard enough to understand all their populations' needs. Their ineffectiveness at reaching underserved groups was contrasted to a more effective targeted strategy and making efforts to build relationships and understand different communities' needs.

"I guess it's going to where they [underserved parent carers] are, rather than them coming to us ... like I said, with the Somali community, we are going out to them ... rather than us just saying, 'yeah, the flyer and whoever comes' we will have to go out there and we'll have to package it differently." (F2, British Bengali mother)

"Going to where they are" was widely reflected as an essential principle. The targeted approach largely depended on providers within the third sector who were representative and/or knowledgeable of their community, with existing contacts. Or it depended on those who were able to develop new relationships with groups to gain access to influential organisations/institutions within specific communities to deliver the message (e.g., community [Somali] group coffee mornings, visual impairment charities, EOTAS support networks, fathers' groups, mosques/masjids). Overall, diverse and informed universal and targeted strategies were advocated to ensure an equitable reach for underserved groups.

# Credibility: establishing trustworthiness of those advertising and/or providing health programmes

Even when parent carers find out about a health programme, participants highlighted that the credibility of the person/organisation ("messenger") was essential. In some cases, this was the programme

<sup>\*\*</sup>Autism Spectrum Condition (ASC)/Attention Deficit Hyperactivity Disorder (ADHD).

<sup>\*\*\*</sup>English Indices of Multiple Deprivation 2019 (44).

TABLE 3 Barriers and strategies to enable participation of underserved parent carer groups in health programmes.

Barriers	Strategies
Reach: To effectively reach	Make efforts to understand and address the needs of diverse communities to ensure an equitable reach for underserved groups.
underserved populations,	Make intentional efforts to communicate with underserved populations who may not be connected to or have awareness of existing
particularly those outside of	networks or places where health programmes are advertised, i.e., use targeted outreach efforts—"Going to where they are."
mainstream networks which	Partner with representative and trusted community organisations that have existing contacts or can develop new relationships within
hinders access to	specific communities to deliver the message.
information and support	Address language and cultural barriers that prevent parent carers from accessing information.
	Recognise the importance of engaging both mothers and fathers and seek appropriate ways to reach and engage both mothers and fathers.
	Use both universal and targeted advertising strategies for the programme.
Credibility: To establish	Recognise the importance of credibility in messaging about health programmes, understanding that mistrust towards messengers or
trustworthiness so that the	organisations can lead to dismissal of the message.
message is not dismissed.	Build trustworthiness through a credible track record of community engagement and by demonstrating an understanding of different
Mistrust towards providers	community needs.
or organizations can lead to	Work with and through trusted community partners and local organizations to promote programmes, tapping into the community's
message dismissal.	existing networks.
	Provide advocacy and practical support to parent carers, demonstrating a commitment to their wellbeing beyond programme promotion.
	Engage in relationship building by spending time with ethnic communities and understanding their specific needs and cultural nuances.
	Utilise word-of-mouth recommendations and testimonials, especially from peers within underserved groups with firsthand experience of
	the programme to enhance credibility.
	Include representative images of parent carers from the target audience in programme advertisements to increase relatability and trust.
Opportunity: There may	Understand that the perception of need might be constrained by sociocultural norms, language, and stigma around seeking support and
be barriers to different	sensitively address these through peer interaction and tailored messaging.
groups' perceptions that a	Present a clear, resonant and appealing message to the target audience, e.g., sensitive use of language, accessible and culturally appropriate
programme meets a relevant	advertising materials.
need	Consider language translations and the choice of wording to make the opportunity more inclusive
1000	Avoid using generic terms like "parent" and instead use "mums and dads."
	Use pictures or videos with fewer words to be more inclusive of parents with dyslexia or those less confident in English.
	Work with underserved communities to highlight the evidential risks to health and mental wellbeing for parent carers.
	Emphasise benefits and attractiveness of the programme to ensure it is perceived as relevant and beneficial.
Reservations: There may	
	<ul> <li>Acknowledge reservations stemming from social anxiety, low confidence, and a sense of isolation among parent carers, which may be greater for underserved groups.</li> </ul>
be barriers to participating	
in the programme, including	Appreciate that acceptance of the parent carer identity may influence openness to joining peer programmes and sense of belonging within
readiness to engage, and social, practical and	the group.  • Create opportunities to meet other parent carers by providing comfortable spaces as a social entrance point, e.g., coffee mornings or groups
emotional factors	
emotional factors	for practical learning, particularly for fathers
	Prioritise pre-programme engagement conversations to build rapport, cultivate readiness, address socioemotional anxieties, assess needs  and also address reading and also address and also address socioemotional anxieties, assess needs  and also address and address and also address and address address and address address and address address and address address and address and address and address and address and addr
	and plan adjustments.
	Address concerns about gender balance within groups and perceptions of fathers' roles  Weak with the attendance of withing balinf structures of other in communities and approximate accordance on the late of the late o
	Work with the strengths of existing belief structures of ethnic communities and empower parent carer awareness of health and
	wellbeing needs.
	Address practical considerations, such as timing, childcare, and transportation, especially for hesitant underserved groups.  Provided the second of th
	Recognise how social circumstances, such as overwhelming caregiving responsibilities and crises, may affect a parents' capacity to be ready
	to participate in a programme. It may be more appropriate for some parents to participate in a future programme.
Optimisation: There may	Understand that mixed-gender or single-gender groups have different implications for attendance and openness and may appeal to
be barriers in the inclusivity,	different groups.
acceptability and	Cultivate a shared social identity around the core common experience(s) of being a parent carer, where attendees feel comfortable regardless
effectiveness of a programme	of other differences.
according to its sensitivity to	Appreciate subtle cultural expectations and preferences, such as communication styles and hospitality.
language, sociocultural	Address the needs of parent carers with sensory impairments or neurodiversity by ensuring materials and activities are accessible.
norms and mode of delivery	Provide translations of programme materials, incorporating cultural adaptations where necessary.
	Consider both online and in-person delivery to accommodate different barriers related to attendance.

provider and in others a signposting organisation, such as the Local Authority. Mistrust towards individuals or organisations providing the message, caused by previous negative interactions, was suggested to create a risk that the "message" would be immediately dismissed.

"... with their association ... they have had a bad experience with this person or that person [e.g., a local authority], and they think this is just an extension to that and it's going to be a similar negative experience..." (A3, British Asian father).

Evidently, the trustworthiness of the messenger was key for people to give initial serious consideration to the message. Such trustworthiness was suggested to be created through a credible track record of authenticity in serving and working in partnership with groups in the local community, with an informed understanding of their culture and needs.

"There's a local organisation, they do health and wellbeing sessions for children, families, carers, the whole community ... if they were promoting such a programme, parents have got trust in the [organisation], so the [organisation] would not promote something that would not be in their best interest. It's about tapping into the local community, and looking at trusted partners that you can work alongside in order to publicise, more than it just being an ad on the internet." (NA11, Black Caribbean British mother)

It was critical how parent carers perceived the intentions that those advertising and providing the programme had for offering it. Best practices were shared of how facilitators within organisations providing the parent carer programme had worked in ways to also become trusted local providers. For example, by providing advocacy and practical support in relation to their child's SEND rights in education or meeting other social needs. Hence, when they informed parent carers about an opportunity, it was received positively and was perceived to be an extension of their support work. Other providers understood similar contextual barriers and engaged in relationship building by spending time with ethnic communities in existing gatherings in order "to let the parents know exactly what the programme is about so they can sign up if they are interested" (F28, Somali mother). Proactive providers also offered events sensitive and appealing to local cultural groups with food and hospitality to draw together those more isolated parent carers, which created opportunities for introduction to the parent carer network. Accordingly, bringing groups of parent carers together was often a foundational precursor to offering programmes.

Another feature which participants emphasised to enhance message credibility was "word of mouth" recommendations. This approach was particularly powerful when imparted by a peer known to the parent, who held an inherent shared empathy with first-hand experience of the programme.

"... because it was recommended by [parent carer from her community] ... I know ... she is a really intelligent lady ... Whatever she's doing, it must be very good ... So I think I would go for it." (NA13, Pakistani mother)

In some situations, previous attendees from particular underserved groups, were tasked with providing honest testimonials to their peers, e.g., via the SEN school. Similarly, adverts with testimonials, including pictures of representative parent carers of the target underserved group (e.g., fathers) was seen as a subtle way to enable resonance, because seeing "people like me" indicated that this programme could be trusted and equally be "relevant for me."

# Opportunity: ensuring that the programme is seen as fulfilling a relevant need

Further potential barriers to access and engagement related to how the opportunity of attending the programme was presented. Participants indicated that the message needed to be clear, resonant and relevant for them individually and within their social group. Therefore, the design of advertising was influential to whether people would perceive an opportunity as relevant and inclusive. Language translations were recommended, and also consideration in the choice of wording was essential. For example, fathers frequently expressed that many would not consider themselves as the primary caregiver and would defer to their partner as the "parent carer," unless the father was a single parent carer, or the established primary caregiver in a couple. Therefore, fathers recommended that the generic "parent" term be avoided when advertising the programme.

"If you then go 'mum and dad', the word 'dad' [is] there so it's not like it feels so exclusive." (A4, White British father, EOTAS)

Using pictures (and fewer words) or videos was suggested as it can be more inclusive of parents with dyslexia or those less confident English speakers.

Many participants articulated a complex barrier regarding whether they might engage with a programme based upon their sense of selfidentity. Participants highlighted how social norms and their social identity could constrain identification of need and support seeking particularly for fathers and some ethnic groups. For some, the "parent carer" term was not familiar as it related to the "SEND world" to which they felt unconnected. Within certain cultural contexts, mothers communicated that they were expected to "just get on with it" irrespective of the challenges they faced in parenting a disabled child. Seeking support for their health needs might be seen as showing weakness implying you are unable to look after your child. Some participants acknowledged that they may prefer "privacy" and to seek what is familiar within their own cultural experience. It was explained that a religious understanding may encourage the person to "pray harder" and seek social support from their "close-knit community" instead of from external programmes. Participants described how reduced knowledge and stigma surrounding childhood disability and mental health, particularly in first generation immigrant populations with low educational status, constrained many parent carers in their community from seeking support. Interpretation of needs could also be restricted by the scope of community languages. For example, it was elucidated that in Somali there are no direct translations for "autism" or "wellbeing," and so parent carers were working within their community to build an understanding through reinterpretation with familiar concepts.

Ultimately, if the advertised programme was to be perceived as relevant, it needed to be seen as a beneficial and attractive opportunity. Further, social norms of "being strong" and "looking after everybody," while potentially seeing aspects of mental wellbeing as "fluffy" or preferring independent self-support, remain in male culture.

"I can care for myself. I can do that on my own. So why do I need someone to tell me about it when I already have ... different sources or different platforms where I can get my information instead of, you know, wasting my time on all these programmes going here and there?" (NA16, Black African father)

Another participant illustrated how intersectionality compounded the issue, highlighting how "a societal issue [exists] in my community about mental health and [being] a man" made "mental wellbeing ... even more of a taboo" and this "misconception about mental health" resulted in a "lack of support for men of our culture" (A3, British Asian father).

Therefore, explaining the evidential risks to health and mental wellbeing for parent carers through peer word of mouth was suggested as a convincing approach to enlighten the relevant need. For participants who had not attended the HPC programme, being distinctively *peer-led* and focused on parent carers themselves stood out as appealing and crucially meeting a gap.

# Reservations: addressing barriers to readiness to participate in health promotion programmes

Even when the need is recognised and opportunity offered, individuals need to be ready and able to engage with the programme. For parent carers to participate, the timing of the programme needs to coincide with the person's readiness to engage. This is shaped by social norms and circumstances.

Social circumstances influenced a person's readiness. Frequently parent carers reported times of being too overwhelmed by their caring responsibilities and in "fighting" for their child's care and education to consider attending a programme for themselves. Providers experienced in working with parent carers with additional social stressors, often in first generation immigrant or other marginalised communities, explained that crises were common, e.g., relating to housing or finances. Therefore, within their hierarchy of needs, parents had less practical and mental/psychological capacity to consider attending a wellbeing programme.

"... some people are just not in the place at that time when things are being run to be able to have the capacity for it ... If they are going through crisis or something is going on where they just cannot focus on anything other than just getting through that ... Where there is just nothing left to give to something, that brain space is just not there. Sometimes that can be a big barrier depending on what else is going on in your life. (F28, Somali mother).

Aware of these social circumstances, some providers had taken steps to meet social needs by providing grocery vouchers at/after each session and indicated that "if they are dropping out, even after that, that shows you how challenging it is for them [to attend]" (F2, British Bengali mother).

Other practical considerations were reported to unequally affect certain parent carers. For working parents, often fathers, the day and time of the group was particularly important. It was pointed out that in some cultures where the father's role was to provide income, the prevailing cost-of-living crisis at the time of this study meant that they were often working longer hours. In some ethnic groups, it was explained that it was uncommon for women to drive and so the venue needed to be local for in-person groups, and partners might need to drive or chaperone them to and from the programme. For single parents, childcare was a major concern for attendance, especially if their child was not in school.

"... immediately I'm thinking about time. I'm a single mum, and I do not do online, so how am I going to do that." (NA22, White British mother with additional needs and a child EOTAS).

Socioemotional reasoning was prominent within parent carers' reservations about attending. Many highlighted social anxiety and low

confidence from their sense of isolation as parent carers, which was accentuated for parent carers of children EOTAS. Understanding and acceptance of the parent carer identity was seen as a critical factor in readiness to attend. Acceptance of the parent carer social identity influenced openness ("is it for me?") to joining with other parent carers in a peer programme. The sense of belongingness within the group was also related to their identity.

"... mums understand each other, we know how we feel, we know... yeah. All from the community." (NA12, Somali single mother).

Fathers were concerned with groups being attended predominantly by mothers: "how am I gonna feel being the only guy there?" (A6, White British father, EOTAS), while worrying about their perceptions of fathers.

"I mean, it can be a bit nerve wracking. It's not that other parents do not treat you as an equal but there is sometimes that kind of misconception that I may not have that maternal instinct, or I may not understand things like how a mum would." (A4, White British father).

Similarly, there were reservations about what such a programme might entail, with fathers concerned it might be "too heavy." Others might feel uncomfortable about sharing, particularly if they were unsure of who was in the group. In seeking to address these barriers and reservations, further efforts were suggested or being undertaken by some providers. In some situations, reservations were simultaneously addressed during word-of-mouth invitations to the programme and other practical support was offered to parent carers, e.g., the EOTAS advocacy group. For some ethnic groups, providers from the same community worked within existing belief structures. In some cases, they explained that deconstruction of unhelpful beliefs around childhood disability and parent mental health was needed to empower parent carers to be self-aware and seek support. Working positively with the strengths of the belief system, the mother's attendance was framed in a way that it would support her to fulfil in their primary caregiving role more effectively.

"I honestly believe the lifestyle that a lot of ethnic minority groups lead are often great and it works great because the man takes the role of being the main income provider and the woman takes the role of looking after the household. Those things work because the community really believes those values so we cannot destruct that. That is the core mentality but at the same time while we are doing those duties we are also looking after ourselves so we can achieve those duties even better. I think that is the approach we have to do." (F36, Bengali mother).

Strategies suggested for engaging fathers included offering practical activities to bring together male parent carers, overcoming stigma, building confidence and developing a gradual openness towards being part of a group of parent carers.

"From what I've seen with the men's group ... it would be such a progressive thing then. Start with a sort of 'Dad-to-Dad' approach, and then say, 'Well, at this point – when you feel comfortable – we are gonna open it up..." (NA29, White British father).

Pre-programme engagement also appeared important to parents for rapport building. Facilitators described how they used pre-programme

conversations with parents as a key mechanism to addressing reservations. The conversation sets a relational tone of "accepting without judgement" while assessing and planning reasonable adjustments in open partnership and allaying anxieties about the programme. This approach is also essential for tailoring programmes to be inclusive.

# Optimisation: tailoring to improve the inclusivity of programme delivery

Several issues were identified for optimising programme delivery to ensure acceptability to the diverse social groups, with varying contexts and preferences.

Group composition was perceived as an important factor. Some mothers from a Muslim background stated that a mixed-gender group would be an immediate barrier to their attendance or would affect their openness if men were present. Some fathers felt that a male-only parent carer group (including facilitators) might "reduce some of the stigma and make them feel a bit more comfortable to open up" (A4, White British father). However, many participants (including fathers and Muslim women) felt there were also significant benefits to learning from others in a diverse group. Further, they believed that the shared parent carer identity and attending "for the same purpose, to access support, provide support, hopefully to learn" was the most important unifying factor for group cohesion.

"It's not about the group has to look like you, or be the same religion, race, or anything else ... you have to feel safe, you have to feel comfortable, you have to feel that you are not going to be judged or critiqued when you are going for support and sharing your experiences. That's not dependent on race." (NA11, Black British Caribbean single mother).

Other subtle cultural expectations were explored that might make people comfortable, for example, in how hospitality was provided. It was also suggested that mothers from some ethnic groups "do not tend to voice" easily in groups and similarly that fathers were more hesitant about being open. Hence the programme design and facilitation approach were seen as critical with dads "looking at more the activities that you do" and not "necessarily start with talking" before feeling safe (NA10, White British father).

Parent carers with sensory impairments or neurodiversity expressed they might feel anxious about attending if they felt that their individual needs were not understood and able to be met. Although those with longstanding visual impairment reported feeling confident asserting their needs up front; one participant (NA34, visually impaired mother) highlighted how important it was for facilitators to "set some time aside ... to make [them] aware of any adjustments" and build a partnership so as they went along they "could figure it out together." Specifically, "workbooks or any kind of materials ... would need the most accessible format" and examples were given where within groups videos were given verbal descriptions or materials were made into large print for parent carers with visual impairment. One participant also suggested it would help "if the materials were developed with the neurodivergent in mind as well as the neurotypical" (A6, neurodivergent White British father of child EOTAS). Facilitators outlined that numerous parent carer attendees on their courses were neurodivergent. This had implications for programme and facilitation adaptation. It was explained that "because of the openness" required on programmes, some autistic parents were "struggling ... with understanding what we are doing" (F31, neurodivergent mother with hearing impairment). Having facilitators with the same lived experience aided these adaptations. Facilitators discussed practices that ensured "no parent carer [was] left behind," "accepting people on their terms" and "suspending judgement" to create the safe space required.

Facilitators' mindfulness of cultural issues helped them to emphasise aspects of programme content that might be more meaningful to some communities. The translation of programme materials into different languages was critical for inclusivity of highly represented populations in one urban area. Translation incorporated cultural reinterpretations where necessary, for example, with the concept of wellbeing and programme activities, to enhance relevance and acceptability in some communities.

There was a reflection across the groups that there was a place for both online and in-person delivery to address different barriers related to access. For example, some parent carers "would not have the luxury to always attend in person"; online delivery might in particular enable programme attendance of parents of children EOTAS or single parents with their child at home. Parent carers with visual impairment perceived fewer barriers online, compared to attending an unknown venue that would require time to learn to navigate. However, it was noted that online delivery also created barriers, particularly relating to digital literacy and poverty. This is relevant to all parent carers but was considered more likely to affect ethnically diverse parents with intersecting factors of social disadvantage. It was notable that two study participants required the support of their PPI chaperone to set up their video call interviews. Many people highly value social gatherings; for example, participants highlighted "that face-to-face forum is really important" for the Somali community.

"...some of them really like face-to-face, they do not like the online stuff to be honest. Yes face-to-face is clearer to be honest, some of them like their English is very hard. They'd rather just face-to-face." (NA12, Somali mother).

Most participants "prefer the face-to-face meetings, training and activities ... rather than the online ones" (NA9, White European mother of EOTAS child).

"I do not think it [online] would be as beneficial. You do not show your emotions in the same way when you are at home. My daughters are always home with me so I keep it together in front of them. I am less willing to be vulnerable." (NA17, White British mother of EOTAS child).

Across different groups, participants reasoned that in-person attendance enabled greater quality social interaction that supported deeper engagement, hence it would be more "memorable" and potentially more impactful.

# Discussion

Our study identifies and builds an understanding of the challenges and barriers that some parent carers might experience to accessing

and engaging with parent carer-focused programmes. Within our UK context, we focused on particular underserved parent carer populations: Global Majority communities, fathers, parents of children EOTAS, those with sensory impairment, and identifying as neurodiverse. The identified challenges and strategies that may help improve the access and engagement of some underrepresented groups of parent carers are summarised (Table 3).

A key challenge is to include parent carers who are not, or less, connected within networks, and who have reduced awareness of childhood disability, SEN and how to access support either for their child or themselves. Cultural factors, language barriers, and social norms further complicate access to such programmes not least because these factors constrain some people's interpretation of their own health needs.

Suggested strategies to improve reach include targeted outreach efforts, community partnerships, and leveraging trusted messengers within underserved communities. The credibility of the messenger advertising a programme is crucial, with trust built through authentic engagement and understanding of community needs. The opportunity for attendance must be presented clearly and resonate with the audience, addressing linguistic, cultural, and social barriers. Strategies to address reservations and cultivating readiness for engaging in a programme include scheduling, addressing socioemotional concerns, and providing practical support. Tailoring delivery for inclusivity involves considering group composition, adapting materials for diverse needs, and alternative options for online and in-person groups. Overall, addressing barriers to access and engagement requires a comprehensive approach, from publicising to programme delivery, tailored to the diverse needs of parent carers in potentially underserved communities. Although we focused on a parent carer programme, similar strategies have been advocated across different types of interventions and populations, indicating that our findings might also be transferrable to other contexts. For example, working with community partners is vital to recognise and align with cultural and religious values in developing sensitive and resonant communication (51). Equally, understanding the specific needs of target populations is critical to create more relevant and impactful health interventions (10).

We acknowledge the broader social determinants of health which affect people's engagement with health behaviours and programmes (6). A social determinants approach to health behaviours seeks to understand how social factors shape people's health reflected by "a dialectic between structure and agency that necessitates situating individuals in context" [p. 78, (52)], which aligns with our findings. For example, our findings highlight deeply rooted sociocultural norms and beliefs particularly in relation to fathers and some ethnic groups. Social norms within these groups were perceived to constrain supportseeking behaviours amongst parent carers (53). In some cases, this may be explained by a hermeneutic injustice, whereby someone's social experience obscures their understanding "owing to prejudicial flaws in shared resources for social interpretation" [p. 147, (54)]. Participants suggested that the sense of stigma linked to parent carer identity that they felt was related to the degree of awareness and acceptance of disability within their own social groups (53, 55–59). Furthermore, mental wellbeing was identified by fathers and some ethnic groups as a "taboo," and an added barrier to seeking support. Some HPC facilitators were aware that if parent carers are to have equal opportunities to access programmes, then some beliefs, attitudes and practices that promote wellbeing to a greater extent need to be supported while constraining beliefs challenged. Providers of parent carer health programmes need to be cognisant of existing belief structures to support families in culturally congruent ways, while also empowering them with knowledge and resources to enable support-seeking behaviours and build sustainable resilience for improved outcomes (53, 60–62). Structural inequalities are likely to require multi-pronged community and policy-level strategies to address inequities beyond our suggested approaches linked to advertising and delivering health programmes (9).

In our study fathers theorised that underrepresentation in parent carer health programmes was related to reduced perceived need or that they preferred to meet their wellbeing needs individually. The level of mental health or wellbeing needs in male parent carers compared to mothers has been less evaluated, although it is acknowledged both parents have different experiences and challenges during the caregiving process (21). A survey in Australia reported that 58% of fathers of disabled children reported high depressive symptoms and 61% stress symptoms, as well as low participation in weekly health-promoting activities, with only 26% undertaking solo exercise and 3% social activity (63). Though this was a small sample, it suggests an unmet need in many fathers. Paradoxically, fathers may be reluctant to attend peer-led parent carer group programmes. Therefore, targeted publicising to address reservations and highlight the potential benefits to fathers participating in a health programme might help increase receptiveness and engagement.

Reducing inequity in access and engagement may require adaptations to intervention implementation and delivery while maintaining intervention functions and change mechanisms (64-66). This was evident in facilitators' views and reported practices in our study. Our study builds on research on parent carer focused programmes by seeking to understand the implementation of such programmes within underserved populations (25). Within the HPC group-based programme, a core intended change mechanism relies on peer support, sharing experiences, and developing a shared social identity of being parent carers who want to improve their health (67). Our findings highlight that personal and social factors (e.g., social disadvantage, gender, ethnicity) might interact with change processes (e.g., feeling safe in the group to share personal experiences). Providers and facilitators need to ensure that these core mechanisms are facilitated when delivering the programme to some underserved groups of parent carers; for example, by drawing on the commonalities of parent carers' experiences and shared identity. Others have pointed to the importance of meaningful engagement within communities, building trust, fostering long-term relationships, often enacted through third sector organisations, to address inequity (5, 7, 68, 69). In our study delivery partner organisations (as providers) were seen to operationalise their understanding of their community context, including social norms, to adapt their strategies to enhance their reach and build trust with underserved communities to enable improved uptake. Further, facilitators with shared lived experience can adapt programme delivery approach and some materials to be more culturally compatible, which adds to previous knowledge (70). Finally, our findings indicate a nuanced debate around equity considerations with access to online health programmes. A recent framework

provides a helpful point of departure for researchers and practitioners to address this challenge at an individual level (71).

Strengths and limitations

We undertook efforts to involve a diverse PPI group and recruit a diverse sample of participants. To ensure that parents from underserved groups were enabled to participate, we worked with key gatekeepers within different communities and partnered with chaperones during interviews. Our analysis and interpretation were challenged and strengthened by the critical contributions and triangulation with the PPI group and other stakeholders.

By interviewing numerous parent carers with no experience and those with experience of parent carer-focused health programmes, we were able to explore a wider variety of perspectives relating to access relevant to the target underserved groups. In addition, parent carers from diverse backgrounds, as facilitators, shared their strategies and successes of promoting and delivering the programme equitably in their community with the authority of lived experience.

Within our sample, participants had multiple intersecting characteristics, so it was not always feasible or sensible to ascribe particular characteristics (e.g., being a father and/or ethnically diverse) directly to specific barriers to access/engagement. Despite some differences in how parent carers with similar demographics might experience inequity, we found that participants described many more common barriers and challenges. Hence our findings were presented for the whole sample with specific examples noted to provide context. While our sample could not be fully exhaustive of all permutations of diversity and intersectionality, we nevertheless did include a range of perspectives from our targeted social groups.

Education is seen as "a fundamental social determinant of health an upstream cause of health" (72). The educational status in our sample, which leant towards more highly educated individuals, may have reduced the diversity of perspectives overall. This meant that certain participants from Global Majority backgrounds, who were second-generation immigrants with higher education, offered their insights on individuals in their communities, typically first-generation immigrants with lower socioeconomic and educational status, often adhering to more "traditional religious" beliefs and norms. There is a danger that assessing individual or family beliefs and practices could become ethnocentric if comparisons are made against untested normative assumptions from researchers without shared cultural experience. While it is impossible to fully understand the lived experience of many of the participants, nevertheless we remained cognisant and empathetic with an open approach that sought to learn from their perspectives and sense-checked these with our PPI and stakeholder groups.

# Conclusion

Addressing barriers to parent carers' equitable access to and engagement with health and wellbeing programmes requires a comprehensive, multifaceted approach, from initial outreach to delivery, tailored to be mindful of the diverse needs of parent carers in our

underserved communities. The identified strategies to help address these barriers, although focused on parent carer programmes, are likely to be applicable to a wider range of health behaviour change interventions.

# Data availability statement

The datasets presented in this article are not readily available because consent was not obtained from participants to share the anonymised data set with other research groups. Requests to access the datasets should be directed to <a href="mailto:christopher.morris@exeter.ac.uk">christopher.morris@exeter.ac.uk</a>.

# **Ethics statement**

This study involving humans were approved by the University of Exeter Medical School Research Ethics Committee (UEMS REC ID: 525009). The studies were conducted in accordance with the local legislation and institutional requirements. The participants provided their written informed consent to participate in this study.

# **Author contributions**

PH: Conceptualization, Data curation, Formal analysis, Investigation, Methodology, Project administration, Validation, Writing - original draft, Writing – review & editing, Visualization. CMc: Writing – original draft, Writing - review & editing, Conceptualization, Data curation, Formal analysis, Investigation, Methodology, Project administration, Validation, Visualization. AM: Conceptualization, Funding acquisition, Project administration, Writing - review & editing, Formal analysis, Validation. FB: Conceptualization, Project administration, Writing - review & editing. AG: Conceptualization, Writing - review & editing, Methodology. SL: Conceptualization, Writing - review & editing. CMo: Conceptualization, Formal analysis, Funding acquisition, Investigation, Methodology, Project administration, Supervision, Validation, Writing - original draft, Writing - review & editing. AB: Conceptualization, Data curation, Formal analysis, Funding acquisition, Investigation, Methodology, Project administration, Supervision, Validation, Writing - original draft, Writing - review & editing.

# Funding

The author(s) declare financial support was received for the research, authorship, and/or publication of this article. This study was funded by the NIHR Programme Grants for Applied Health Research through a Programme Development Grant (NIHR204583), and supported by the NIHR ARC Southwest Peninsula (known as PenARC). The views expressed are those of the authors and not necessarily those of the NIHR or the Department of Health and Social Care.

# Acknowledgments

We thank members of the PenCRU Family Faculty group for their guidance throughout this project. We would especially like to thank

the participants for finding time in their busy schedules to share their thoughts and experiences.

Conflict of interest

The authors declare that the research was conducted in the absence of any commercial or financial relationships that could be construed as a potential conflict of interest.

# Publisher's note

All claims expressed in this article are solely those of the authors and do not necessarily represent those of their affiliated organizations, or those of the publisher, the editors and the reviewers. Any product that may be evaluated in this article, or claim that may be made by its manufacturer, is not guaranteed or endorsed by the publisher.

# **Author disclaimer**

The views expressed are those of the authors and not necessarily those of the NIHR or the Department of Health and Social Care.

# Supplementary material

The Supplementary material for this article can be found online at: https://www.frontiersin.org/articles/10.3389/fpubh.2024.1445879/full#supplementary-material

# References

- 1. Wight D, Wimbush E, Jepson R, Doi L. Six steps in quality intervention development (6SQuID). *J Epidemiol Community Health*. (2016) 70:520–5. doi: 10.1136/jech-2015-205952
- 2. Veinot TC, Mitchell H, Ancker JS. Good intentions are not enough: how informatics interventions can worsen inequality. *J Am Med Inform Assoc.* (2018) 25:1080–8. doi: 10.1093/jamia/ocy052
- Rothman AJ, Simpson JA, Huelsnitz CO, Jones RE, Scholz U. Integrating intrapersonal and interpersonal processes: a key step in advancing the science of behavior change. Health Psychol Rev. (2020) 14:182–7. doi: 10.1080/17437199. 2020.1719183
- $4.\ Michie S, Atkins L, West R, Eds.\ The behaviour change wheel: a guide to designing interventions.\ London: Silverback Publishing. (2014).$
- 5. Marmot M. Fair society, healthy lives: a strategic review of inequalities in England. London: University College London (2010).
- 6. Alcántara C, Diaz SV, Cosenzo LG, Loucks EB, Penedo FJ, Williams NJ. Social determinants as moderators of the effectiveness of health behavior change interventions: scientific gaps and opportunities. *Health Psychol Rev.* (2020) 14:132–44. doi: 10.1080/17437199.2020.1718527
- 7. Rai T, Hinton L, McManus RJ, Pope C. What would it take to meaningfully attend to ethnicity and race in health research? Learning from a trial intervention development study. *Sociol Health Illn.* (2022) 44:57–72. doi: 10.1111/1467-9566.13431
- 8. Rai T, Dixon S, Ziebland S. Shifting research culture to address the mismatch between where trials recruit and where populations with the most disease live: a qualitative study. *BMC Med Res Methodol.* (2021) 21:80. doi: 10.1186/s12874-021-01268-z
- 9. Mattaini MA, Roose KM, Fawcett SB. Behavioral interventions contributing to reducing poverty and inequities. *Behav Soc Issues.* (2023) 32:314–37. doi: 10.1007/s42822-022-00114-1
- 10. Davidson EM, Liu JJ, Bhopal RAJ, White M, Johnson MRD, Netto G, et al. Behavior change interventions to improve the health of racial and ethnic minority populations: a tool kit of adaptation approaches. *Milbank Q*. (2013) 91:811–51. doi: 10.1111/1468-0009.12034
- 11. Hoyer D, Dee E, O'Leary MS, Heffernan M, Gelfand K, Kappel R, et al. How do we define and measure health equity? The state of current practice and tools to advance health equity. *J Public Health Manag Pract.* (2022) 28:570–7. doi: 10.1097/PHH.000000000001603
- 12. Tettey W. Inspiring inclusive excellence-professor wisdom tettey's installation address. Scarborough, Canada: University of Toronto Scarborough, (2019). Available at: https://www.utsc.utoronto.ca/news-events/inspiring-inclusive-excellence-professor-wisdom-tetteys-installation-address (Accessed May 24, 2024).
- 13. The National Institute of Health and Care Research (NIHR). Under-served communities. (2024). Available at: https://www.nihr.ac.uk/about-us/our-key-priorities/under-served-communities.htm (Accessed February 23, 2024).
- 14. Masefield SC, Prady SL, Sheldon TA, Small N, Jarvis S, Pickett KE. The caregiver health effects of caring for young children with developmental disabilities: a Meta-analysis. *Matern Child Health J.* (2020) 24:561–74. doi: 10.1007/s10995-020-02896-5
- 15. Lach LM, Kohen DE, Garner RE, Brehaut JC, Miller AR, Klassen AF, et al. The health and psychosocial functioning of caregivers of children with neurodevelopmental disorders. *Disabil Rehabil.* (2009) 31:607–18. Available from: doi: 10.1080/0963828 0802242163

- 16. Rydzewska E, Dunn K, Cooper S-A, Kinnear D. Mental ill-health in mothers of people with intellectual disabilities compared with mothers of typically developing people: a systematic review and meta-analysis. *J Intellect Disabil Res.* (2021) 65:501–34. doi: 10.1111/jir.12827
- 17. Hayes SA, Watson SL. The impact of parenting stress: a Meta-analysis of studies comparing the experience of parenting stress in parents of children with and without autism Spectrum disorder. *J Autism Dev Disord.* (2013) 43:629–42. doi: 10.1007/s10803-012-1604-y
- 18. Okoli ML, Ogbu CE, Enyi CO, Okoli IC, Wilson RE, Kirby RS. Sociodemographic and socioeconomic correlates of learning disability in preterm children in the United States. *BMC Public Health*. (2022) 22:212. doi: 10.1186/s12889-022-12592-4
- 19. Spencer NJ, Blackburn CM, Read JM. Disabling chronic conditions in childhood and socioeconomic disadvantage: a systematic review and meta-analyses of observational studies. *BMJ Open.* (2015) 5:e007062. doi: 10.1136/bmjopen-2014-007062
- 20. Carmassi C, Dell'Oste V, Foghi C, Bertelloni CA, Conti E, Calderoni S, et al. Post-traumatic stress reactions in caregivers of children and adolescents/young adults with severe diseases: a systematic review of risk and protective factors. *Int J Environ Res Public Health.* (2020) 18:189. doi: 10.3390/ijerph18010189
- 21. Hatzmann J, Maurice-Stam H, Heymans HS, Grootenhuis MA. A predictive model of health related quality of life of parents of chronically ill children: the importance of care-dependency of their child and their support system. *Health Qual Life Outcomes*. (2009) 7:72. doi: 10.1186/1477-7525-7-72
- 22. Berghs M, Atkin K, Graham H, Hatton C, Thomas C. Public Health Research. Implications for public health research of models and theories of disability: a scoping study and evidence synthesis. Southampton (UK): NIHR Journals Library Copyright © (2016).
- 23. Laverack G. Health promotion practice: building empowered communities. Maidenhead, UK: Open University Press: McGraw-Hill Education (2007).
- 24. Borek AJ, McDonald B, Fredlund M, Bjornstad G, Logan S, Morris C. Healthy parent carers programme: development and feasibility of a novel group-based health-promotion intervention. *BMC Public Health*. (2018) 18:270. doi: 10.1186/s12889-018-5168-4
- 25. Bourke-Taylor HM, Jane F, Peat J. Healthy mothers healthy families workshop intervention: a preliminary investigation of healthy lifestyle changes for mothers of a child with a disability. *J Autism Dev Disord*. (2019) 49:935–49. doi: 10.1007/s10803-018-3789-1
- 26. Schlebusch L, Chambers N, Rosenstein D, Erasmus P, de Vries PJ. Supporting caregivers of children with developmental disabilities: findings from a brief caregiver well-being programme in South Africa. *Autism.* (2024) 28:199–214. doi: 10.1177/13623613221133182
- 27. Lancaster K, Bhopti A, Kern ML, Taylor R, Janson A, Harding K. Effectiveness of peer support programmes for improving well-being and quality of life in parents/carers of children with disability or chronic illness: a systematic review. *Child Care Health Dev.* (2023) 49:485–96. doi: 10.1111/cch.13063
- 28. Lloyd J, Bjornstad G, Borek A, Cuffe-Fuller B, Fredlund M, McDonald A, et al. Healthy parent carers programme: mixed methods process evaluation and refinement of a health promotion intervention. *BMJ Open.* (2021) 11:e045570. doi: 10.1136/bmjopen-2020-045570
- 29. Bjornstad G, Cuffe-Fuller B, Ukoumunne OC, Fredlund M, McDonald A, Wilkinson K, et al. Healthy parent carers: feasibility randomised controlled trial of a

peer-led group-based health promotion intervention for parent carers of disabled children. *Pilot Feasibil. Stud.* (2021) 7:144. doi: 10.1186/s40814-021-00881-5

- 30. Bjornstad G, Wilkinson K, Cuffe-Fuller B, Fitzpatrick K, Borek A, Ukoumunne OC, et al. Healthy parent carers peer-led group-based health promotion intervention for parent carers of disabled children: protocol for a feasibility study using a parallel group randomised controlled trial design. *Pilot Feasibil. Stud.* (2019) 5:137. doi: 10.1186/s40814-019-0517-3
- 31. Corry O. Defining and theorizing the third sector In: R Taylor, editor. Third sector research. New York, NY: Springer (2010). 11–20.
- 32. Shakespeare T, Watson N. The social model of disability: an outdated ideology? In: NB Sharon and MA Barbara, editors. Exploring theories and expanding methodologies: where we are and where we need to go. Research in social science and disability. 2. Leeds: Emerald Group Publishing Limited (2001). 9–28.
- 33. Colver A. A shared framework and language for childhood disability. Dev. Med. Child Neurol. (2005) 47:780–4. doi: 10.1111/j.1469-8749.2005.tb01078.x
- 34. Children Act. HM Government, UK. The Children and Families Act 2014, UK Public General Acts 2014 c. 6. (2004). Available at: https://www.legislation.gov.uk/ukpga/2014/6/contents (Accessed July 06, 2024).
- 35. Special Educational Needs and Disability Act. HM Government, UK. The Children and Families Act 2014, UK. Part 3: Children and young people in England with special educational needs or disabilities. Public General Acts 2014 c. 6. (2001). Available at: <a href="https://www.legislation.gov.uk/ukpga/2014/6/contents">https://www.legislation.gov.uk/ukpga/2014/6/contents</a> (Accessed July 06, 2024).
- 36. Campbell-Stephens RM. Introduction: global majority decolonising narratives. Educational leadership and the global majority: decolonising narratives. Cham: Springer International Publishing. (2021). 1–21.
- 37. Crenshaw KW. Demarginalizing the intersection of race and sex: a black feminist critique of antidiscrimination doctrine, feminist theory and antiracist politics In: AM Jaggar, editor. Living with contradictions. *1st* ed. New York: Routledge (1994)
- 38. Bauer GR. Incorporating intersectionality theory into population health research methodology: challenges and the potential to advance health equity. *Soc Sci Med.* (2014) 110:10–7. doi: 10.1016/j.socscimed.2014.03.022
- 39. Pen CRU, Faculty F. Pen CRU family faculty handbook. (2021). Available at: https://www.pencru.org/media/universityofexeter/medicalschool/subsites/pencru/PenCRU\_and\_Family\_Faculty\_Handbook\_2021.pdf (Accessed February 09, 2024).
- 40. Staniszewska S, Brett J, Simera I, Seers K, Mockford C, Goodlad S, et al. GRIPP2 reporting checklists: tools to improve reporting of patient and public involvement in research. *BMJ*. (2017) 358;j3453. doi: 10.1136/bmj.j3453
- 41. NHS\_England. Core 20PLUS5 (adults) an approach to reducing healthcare inequalities. (2023). Available at: https://www.nglandnhsuk/about/equality/equality-hub/national-healthcare-inequalities-improvement-programme/core20plus5/ (Accessed April 03, 2024).
- 42. McCluskey G, Riddell S, Weedon E. Children's rights, school exclusion and alternative educational provision. *Int J Incl Educ.* (2015) 19:595–607. doi: 10.1080/13603116.2014.961677
- 43. Paulauskaite L, Timmerman A, Kouroupa A, Allard A, Gray KM, Hastings RP, et al. Elective home education of children with neurodevelopmental conditions before and after the COVID-19 pandemic started. *Front Psychol.* (2022) 13:995217. doi: 10.3389/fpsyg.2022.995217
  - 44. Statistics OfN. English indices of deprivation 2019. (2019).
- 45. Ronzi S, Gravenhorst K, Rinaldi C, Villarroel-Williams N, Ejegi-Memeh S, McGowan VJ, et al. Intersectionality and public understandings of health inequity in England: learning from the views and experiences of young adults. *Public Health*. (2023) 222:147–53. doi: 10.1016/j.puhe.2023.07.002
- 46. Breckenridge J, Jones D. Demystifying theoretical sampling in grounded theory research. *Grounded Theory Rev.* (2009) 8:113–26.
- 47. Agarwal A, Han J, Goldberg D. Effective use of medical chaperones: practical strategies and perspectives. JAMA Dermatol. (2023) 159:471–2. doi: 10.1001/jamadermatol.2023.0185
- 48. Braun V, Clarke V. Using the matic analysis in psychology.  $\it Qual\,Res\,Psychol.$  (2006) 3:77–101. doi: 10.1191/1478088706qp063oa
- 49. Morse JM. Critical analysis of strategies for determining rigor in qualitative inquiry.  $Qual\ Health\ Res.\ (2015)\ 25:1212-22.\ doi: 10.1177/1049732315588501$
- 50. Birt L, Scott S, Cavers D, Campbell C, Walter F. Member checking: a tool to enhance trustworthiness or merely a nod to validation? *Qual Health Res.* (2016) 26:1802–11. doi: 10.1177/1049732316654870

- 51. Netto G, Bhopal R, Lederle N, Khatoon J, Jackson A. How can health promotion interventions be adapted for minority ethnic communities? Five principles for guiding the development of behavioural interventions. *Health Promot Int.* (2010) 25:248–57. doi: 10.1093/heapro/daq012
- 52. Short SE, Mollborn S. Social determinants and health behaviors: conceptual frames and empirical advances. *Curr Opin Psychol.* (2015) 5:78–84. doi: 10.1016/j. copsyc.2015.05.002
- 53. Croot EJ, Grant G, Cooper CL, Mathers N. Perceptions of the causes of childhood disability among Pakistani families living in the UK. *Health Soc Care Community*. (2008) 16:606–13. doi: 10.1111/j.1365-2524.2008.00784.x
- 54. Fricker M. Epistemic injustice: power and the ethics of knowing Oxford. Oxford: Oxford University Press (2007).
- 55. Abdul-Chani MM, Moreno CP, Reeder JA, Zuckerman KE, Lindly OJ. Perceived community disability stigma in multicultural, low-income populations: measure development and validation. *Res Dev Disabil.* (2021) 115:103997. doi: 10.1016/j. ridd.2021.103997
- 56. Tekola B, Kinfe M, Girma F, Hanlon C, Hoekstra RA. Perceptions and experiences of stigma among parents of children with developmental disorders in Ethiopia: a qualitative study. Soc Sci Med. (2020) 256:113034. doi: 10.1016/j.socscimed.2020.113034
- 57. Song J, Mailick MR, Greenberg JS. Health of parents of individuals with developmental disorders or mental health problems: impacts of stigma. *Soc Sci Med.* (2018) 217:152–8. doi: 10.1016/j.socscimed.2018.09.044
- 58. Pelchat D, Lefebvre H, Perreault M. Differences and similarities between mothers' and fathers' experiences of parenting a child with a disability. *J Child Health Care.* (2003) 7:231–47. doi: 10.1177/13674935030074001
- 59. Sim SS, Bourke-Taylor H, Fossey E, Yu M-l. Being more than a mother: a qualitative study of Asian immigrant mothers in Australia who have children with disabilities. *Res Dev Disabil.* (2021) 117:104060. doi: 10.1016/j.ridd.2021.104060
- 60. Van Zyl LE, Stander MW. A strengths-based approach towards coaching in a multicultural environment In: JHD Cornelius-White, R Motschnig-Pitrik and M Lux, editors. Interdisciplinary handbook of the person-centered approach: research and theory. New York, NY: Springer New York (2013). 245–57.
- 61. Whiting M, Nash AS, Kendall S, Roberts SA. Enhancing resilience and self-efficacy in the parents of children with disabilities and complex health needs. *Prim Health Care Res Dev.* (2019) 20:e33. doi: 10.1017/S1463423619000112
- 62. Bourke-Taylor HM, Joyce KS, Grzegorczyn S, Tirlea L. Profile of mothers of children with a disability who seek support for mental health and wellbeing. *J Autism Dev Disord*. (2022) 52:3800–13. doi: 10.1007/s10803-021-05260-w
- 63. Bourke-Taylor HM, Cotter C, Joyce KS, Reddihough DS, Brown T. Fathers of children with a disability: health, work, and family life issues. *Disabil Rehabil.* (2022) 44:4441–51. doi: 10.1080/09638288.2021.1910739
- $64.\,Graham\,H,$  Kelly M. Health inequalities: concepts, frameworks and policy. Briefing Paper. NHS Health Development Agency (UK). (2004).
- 65. Lorenc T, Petticrew M, Welch V, Tugwell P. What types of interventions generate inequalities? Evidence from systematic reviews. *J Epidemiol Community Health*. (2013) 67:190–3. doi: 10.1136/jech-2012-201257
- 66. Moore G, Campbell M, Copeland L, Craig P, Movsisyan A, Hoddinott P, et al. Adapting interventions to new contexts—the ADAPT guidance. *BMJ*. (2021) 374:n1679. doi: 10.1136/bmj.n1679
- 67. Borek AJ, Abraham C, Greaves CJ, Gillison F, Tarrant M, Morgan-Trimmer S, et al. Identifying change processes in group-based health behaviour-change interventions: development of the mechanisms of action in group-based interventions (MAGI) framework. *Health Psychol Rev.* (2019) 13:227–47. doi: 10.1080/17437199.2019.1625282
- 68. Bach-Mortensen AM, Montgomery P. What are the barriers and facilitators for third sector organisations (non-profits) to evaluate their services? A systematic review. *Syst Rev.* (2018) 7:13. doi: 10.1186/s13643-018-0681-1
- 69. Zasada M. Entrepreneurial activity in community health promotion organisations: findings from an ethnographic study. *Soc Enterp J.* (2017) 13:144–62. doi: 10.1108/SEJ-07-2016-0030
- 70. Garrood A, Bjornstad G, Borek A, Gillett A, Lloyd J, Brand S, et al. Healthy parent carers: acceptability and practicability of online delivery and learning through implementation by delivery partner organisations. *Health Expect.* (2023) 26:2050–63. doi: 10.1111/hex.13812
- 71. Miller SJ, Sly JR, Alcaraz KI, Ashing K, Christy SM, Gonzalez B, et al. Equity and behavioral digital health interventions: strategies to improve benefit and reach. *Transl Behav Med.* (2023) 13:400–5. doi: 10.1093/tbm/ibad010
- 72. Hahn RA, Truman BI. Education improves public health and promotes health equity. *Int J Health Serv.* (2015) 45:657–78. doi: 10.1177/0020731415585986